Author's response to reviews

Title: Tanzanian malignant lymphomas: WHO classification, presentation, ploidy, proliferation and HIV/EBV association

Authors:

Amos R Mwakigonja (amwakigonja@muhas.ac.tz)
Ephata E Kaaya (ekaaya@muhas.ac.tz)
Thomas Heiden (Thomas.Heiden@charite.de)
German Wannhoff (wannhoff@gmail.com)
Juan Castro (Juan.Castro@ki.se)
Fatemeh Pak (parisa_amin@yahoo.com)
Anna Porwit (anna.porwit@karolinska.se)
Peter Biberfeld (Peter.Biberfeld@ki.se)

Version: 9 Date: 26 June 2010

Author's response to reviews:

Dear Editors,

We are very grateful for accepting our article for publication into BMC Cancer. We are also very grateful for all the formatting and corrections you have given us. By and large we have done proofing and attended all typographicals that we could find. However, we remain available to any further, last minute corrections and proofing. The latest changes are in track changes format. Kindly accept them after approval.

Sincerely yours,

Amos Mwakigonja,

Corresponding author.